Author's response to reviews

Title: Chemotherapy Followed by Surgery versus Surgery Alone in Patients with Resectable Oesophageal Squamous Cell Carcinoma: Long-term Results of a Randomized Controlled Trial

Authors:

Jurjen J Boonstra (j.boonstra1@antoniusziekenhuis.nl)
Tjebbe C Kok (KokT@maasstadziekenhuis.nl)
Bas PL Wijnhoven (b.wijnhoven@erasmusmc.nl)
Mark van Heijl (m.vanheijl@amc.uva.nl)
Mark I van Berge Henegouwen (M.I.vanBergeHenegouwen@amc.uva.nl)
Fiebo JW ten Kate (F.J.W.tenKate@umcutrecht.nl)
Peter D Siersema (P.D.Siersema@umcutrecht.nl)
Winand NM Dinjens (w.dinjens@erasmusmc.nl)
Jan JB van Lanschot (j.vanlanschot@erasmusmc.nl)
Hugo W Tilanus (h.w.tilanus@erasmusmc.nl)
Ate van der Gaast (a.vandergaast@erasmusmc.nl)

Version: 5  Date: 19 May 2011

Author's response to reviews: see over
May 19, 2011

Dear Mr. Cochrane,

Please find enclosed the revised version (Version 2) of manuscript MS 9957380074272586, entitled “Chemotherapy Followed by Surgery versus Surgery Alone in Patients with Resectable Oesophageal Squamous Cell Carcinoma: Long-term Results of a Randomized Controlled Trial” by Jurjen J. Boonstra, Tjebbe C. Kok, Bas P.L. Wijnhoven, Mark van Heijl, Mark I. van Berge Henegouwen, Fiebo J. W. ten Kate, Peter D. Siersema, Winand N.M. Dinjens, Jan J.B. van Lanschot, Hugo W. Tilanus, Ate van der Gaast.

We would like to thank the editors for valuable suggestions for further improvement of the manuscript.

Small typographical errors on pages 14 to 17 have been corrected.

The authors' contributions paragraph is formatted as a paragraph. All authors read and approved the final version of the manuscript.

On behalf of all authors,

Yours Sincerely,

Jurjen J Boonstra, MD
Department of Surgery/Pathology
Josephine Nefkens Institute, room Be-332
Erasmus University Medical Center
P.O. Box 2040
3000 CA Rotterdam
The Netherlands
Phone: +31-10-7044404
Fax: +31-10-7044762
E-mail: j.boonstra1@antoniusziekenhuis.nl